\documentclass{beamer}
%\documentclass[t,handout]{beamer}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\usepackage{amsmath,amssymb}
\usepackage{beamerthemesplit}
\usepackage{graphicx, amsfonts}
\usetheme[secheader]{Boadilla}

\mode<presentation>
\title[My Talk]{Probability Distributions}

\author[DRHundley]{Douglas Hundley}
\institute[WHI]{Whitman College}
\date[04/01/2013]{Spring 2013}

\begin{document}
\frame{\titlepage}

\section[Introduction]{Definitions}

\begin{frame}

 Here is Euler's Identity: $$e^{i\theta} = \cos\theta + i \sin\theta.$$
 (one of my favorites)
 \pause

 Notice that when $\theta = \pi$:
 \begin{eqnarray*}
  e^{i\pi} & = & \cos\pi + i \sin\pi\\
  e^{i\pi} & = & -1 \\
  e^{i\pi}+1 & = & 0 
 \end{eqnarray*}



\end{frame}

\begin{frame}[fragile]
 
 Here is an example of using the \verb+verbatim+ command:
\begin{verbatim}
plot(sin(x),x=-Pi..Pi);
\end{verbatim} 

\end{frame}

\section[Blocks]{Alert Blocks}
\begin{frame}
\begin{alertblock}{An Alert Block}
This is an alert block
\end{alertblock}

\begin{block}{A block}
A regular block
\end{block}
\end{frame}


\end{document}


